Author's response to reviews

Title: Development of acute pancreatitis caused by sodium valproate in a patient with bipolar disorder on hemodialysis for chronic renal failure: a case report

Authors:

Hiroaki Okayasu (hokayasu@dokkyomed.ac.jp)
Takahiro Shinozaki (tshinoz@dokkyomed.ac.jp)
Akira Osone (osone@dokkyomed.ac.jp)
Yuji Ozeki (ozeki@dokkyomed.ac.jp)
Kazutaka Shimoda (shimoda@dokkyomed.ac.jp)

Version: 11 Date: 22 March 2014

Author's response to reviews:

Professor Alice Murray,
Executive Editor
BMC psychiatry
March 20, 2014

Dear Professor Alice Murray,

Subject: Submission of revised manuscript entitled “Development of acute pancreatitis caused by sodium valproate in a patient with bipolar disorder on hemodialysis for chronic renal failure” (MS: 1542735866116522)

Thank you very much for your letter via e-mail dated March 18, 2014 with regard to our manuscript, together with the comment. We are very grateful for your valuable time in reviewing our manuscript.

I am sending herewith the revised version in accordance with the comment of the reviewer. Our response is given below.

We hope that the revised manuscript is now acceptable for publication.
We look forward to your favorable response.

To Editor

A list of figure legends have been provided at the end of the manuscript after the 'references' section whilst the figures have been uploaded as separate files during the submission process.

Yours sincerely,

Hiroaki Okayasu, MD, PhD
Department of Psychiatry,
Dokkyo Medical University,
880 Kitakobayashi
Mibu, 321-0293
JAPAN
Tel: +81-282-86-1111
Fax: +81-282-86-5187
E-mail: hokayasu@dokkyomed.ac.jp